

Citation: Coelho AC, Oliveira JC, Espada CR, Reimão JQ, Trinconi CT, Uliana SRB (2016) A Luciferase-Expressing *Leishmania braziliensis* Line That Leads to Sustained Skin Lesions in BALB/c Mice and Allows Monitoring of Miltefosine Treatment Outcome. PLoS Negl Trop Dis 10(5): e0004660. doi:10.1371/journal.pntd.0004660

Editor: Genevieve Milon, Institut Pasteur, FRANCE

Received: December 31, 2015

Accepted: April 2, 2016

Published: May 4, 2016

Copyright: © 2016 Coelho et al. This is an open access article distributed under the terms of the <u>Creative Commons Attribution License</u>, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All relevant data are within the paper and its Supporting Information files.

Funding: This work was supported by research grants from Fundação de Amparo à Pesquisa do Estado de São Paulo (FAPESP, 2011/20484-7 and 2015/09080-2) and Conselho Nacional de Desenvolvimento Científico e Tecnológico (CNPq, 473343/2012-6), Brazil. SRBU is the recipient of a senior researcher scholarship from CNPq. ACC, CRE, JQR and CTT were supported by FAPESP (2012/14629-5, 2015/05130-5, 2011/219702 and 2011/18858-6, respectively); JCO was supported by **RESEARCH ARTICLE**

A Luciferase-Expressing *Leishmania braziliensis* Line That Leads to Sustained Skin Lesions in BALB/c Mice and Allows Monitoring of Miltefosine Treatment Outcome

Adriano C. Coelho^{¤a}, Jordana C. Oliveira, Caroline R. Espada, Juliana Q. Reimão^{¤b}, Cristiana T. Trinconi, Silvia R. B. Uliana*

Departamento de Parasitologia, Instituto de Ciências Biomédicas, Universidade de São Paulo, São Paulo, Brazil

¤a Current address: Departamento de Biologia Animal, Instituto de Biologia, Universidade Estadual de Campinas, Campinas, Brazil

¤b Current address: Departamento de Morfologia e Patologia Básica, Faculdade de Medicina de Jundiaí, Jundiai, São Paulo, Brazil

* srbulian@icb.usp.br

Abstract

Background

Leishmania braziliensis is the most prevalent species isolated from patients displaying cutaneous and muco-cutaneous leishmaniasis in South America. However, there are difficulties for studying *L. braziliensis* pathogenesis or response to chemotherapy *in vivo* due to the natural resistance of most mouse strains to infection with these parasites. The aim of this work was to develop an experimental set up that could be used to assess drug efficacy against *L. braziliensis*. The model was tested using miltefosine.

Methodology/Principal Findings

A *L*. *braziliensis* line, originally isolated from a cutaneous leishmaniasis patient, was passaged repeatedly in laboratory rodents and further genetically manipulated to express luciferase. Once collected from a culture of parasites freshly transformed from amastigotes, 10⁶ wild type or luciferase-expressing stationary phase promastigotes were inoculated subcutaneously in young BALB/c mice or golden hamsters. In both groups, sustained cutaneous lesions developed at the site of inoculation, no spontaneous self- healing being observed 4 months post-inoculation, if left untreated. Compared to the wild type line features, no difference was noted for the luciferase-transgenic line. Infected animals were treated with 5 or 15 mg/kg/day miltefosine orally for 15 days. At the end of treatment, lesions had regressed and parasites were not detected. However, relapses were observed in animals treated with both doses of miltefosine. Coordenação de Aperfeiçoamento de Pessoal de Ensino Superior (CAPES). The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing Interests: The authors have declared that no competing interests exist.

Conclusions/Significance

Here we described experimental settings for a late-healing model of cutaneous leishmaniasis upon inoculation of a luciferase-expressing *L*. *braziliensis* line that can be applied to drug development projects. These settings allowed the monitoring of the transient efficacy of a short-term miltefosine administration.

Author Summary

Leishmania braziliensis is the most prevalent species isolated from patients displaying either cutaneous or mucocutaneous leishmaniasis in South America. In this study, we developed a transgenic luciferase-expressing *L. braziliensis* line. These parasites were passaged in hamsters and mice and then transformed back into promastigotes. Once inoculated subcutaneously in the footpad of young laboratory animals—BALB/c mice or golden hamsters, rapid and sustained footpad thickness increase developed. This experimental model was used to monitor the parasite load fluctuations and the response to miltefosine treatment. Mice were treated orally over a two-week period, starting at week 4 post-inoculation. Though such a regimen was shown to display efficacy, the effect was not sustained and both parasite re-expansion and delayed footpad thickness increase were noticed.

Introduction

Leishmania spp. are the etiological agents of leishmaniasis, a complex of vector-borne infectious diseases transmitted by sand flies. The disease is widespread in the world in tropical and subtropical regions. Approximately 12 million people are affected worldwide and 1.2 million new cases occur each year [1]. Leishmaniasis is responsible for a spectrum of clinical manifestations including visceral, cutaneous, mucosal, disseminated and diffuse cutaneous disease that are determined, at least partially, by the species of the parasite [2]. In South America, *Leishmania (Viannia) braziliensis* is the most prevalent species isolated from patients displaying either cutaneous or mucocutaneous leishmaniasis [3-5]. This species is also associated with disseminated leishmaniasis [5].

In spite of its epidemiological importance, the study of *L. braziliensis* infections is limited by the difficulty of establishing good animal models of disease. *L. braziliensis* infects hamsters and induces a chronic cutaneous disease with no spontaneous healing [6, 7]. However, inbred strains of mice are generally resistant to *L. braziliensis* or develop a transient and self-healing cutaneous lesion [7-10]. When the parasites are inoculated into partially susceptible strains such as BALB/c mice, a non-ulcerative lesion develops initially and evolves to spontaneous healing in about 30 days [8, 10]. A mouse model of sustained skin lesions with *L. braziliensis* would be very useful for studies on the pathogenesis of the disease as well as a tool to develop new control strategies, such as vaccines and drugs.

The treatment of leishmaniasis relies on a few drugs, most of them displaying some inadequacy to present requirements. They are expensive, toxic and require prolonged and parenteral administration. Additionally, drug resistance has been reported [11, 12]. More than a decade ago, miltefosine (MF) was described as an oral drug against visceral leishmaniasis (VL) and has been widely used in the Indian subcontinent for that application [13–15]. On the other hand, MF efficacy for the treatment of cutaneous leishmaniasis (CL) seems to be dependent on the *Leishmania* species and even on intraspecies heterogeneity. Some clinical studies in South America identified cure rates varying from 53% to 91% in CL patients $[\underline{16-18}]$. Two clinical studies performed in Brazil detected cure rates of 71.4% and 75%, respectively, in infections due to *L. guyanensis* [<u>19</u>] and *L. braziliensis* [<u>20</u>]. However, given the lack of other new treatment alternatives, further understanding of MF's potentialities and limitations is necessary.

We have recently described the use of luciferase as a reporter for drug efficacy evaluation in cutaneous and visceral leishmaniasis models [21, 22]. In both models, luciferase has proven to be an accurate measure of parasite burden.

In this study, we generated a transgenic line of *L. braziliensis* expressing luciferase that was adapted to induce chronic lesions in BALB/c mice. Using this experimental model, we evaluated the efficacy of MF to reduce lesion size and parasite load.

Methods

Drugs

Amphotericin B sodium deoxycholate and MF were obtained from Sigma-Aldrich (St. Louis, MO, USA). Stock solutions of amphotericin B and MF for *in vitro* experiments were prepared in sterile water (10 mM). Miltefosine used for the treatment of infected mice was prepared daily in sterile saline (0.9% NaCl) from a 6 mg/mL stock solution.

Parasites

The *L. braziliensis* strain MHOM/BR/94/H3227 was kindly provided by Dr. Maria Jania Teixeira, from Universidade Federal do Ceará. These parasites were isolated from a cutaneous ulcer from a patient in Ceará State, Brazil and the strain was previously typed as *L. braziliensis* [10]. Promastigotes were grown in medium M199 (Sigma-Aldrich) supplemented with 10% heat-inactivated foetal calf serum, 2% urine, 0.25% hemin, 12 mM NaHCO₃, 50 U/mL penicillin and 50 µg/mL streptomycin at 25°C. For *in vivo* experiments, female BALB/c mice (4–5 weekold) were inoculated with 10⁶ stationary-phase promastigotes injected subcutaneously in the right hind footpad. Amastigotes were obtained from infected mice, as described [23] and differentiated back as promastigotes in M199. Promastigotes were counted in a Neubauer haemocytometer.

Generation of a transgenic line of L. braziliensis expressing luciferase

A construct containing the luc2 gene was previously built [22] taking advantage of the vectors pSSU-int and pSPαHYGα which were kindly provided by Dr. Tony Aebischer (Robert Koch Institute, Berlin, Germany) and Dr. Marc Ouellette (Universite Laval, Quebec, Canada). The linear cassette purified upon Pac I and Pme I digestion of the construct described previously, was used to transfect L. braziliensis H3227 promastigotes. Briefly, the cassette contains the luc2 gene followed by a Leishmania 3' UTR, the hygromycin phosphotransferase gene and fragments of the L. mexicana small subunit (SSU) ribosomal DNA (rDNA) at the cassette extremities to promote homologous recombination. Transfection was performed as described [24] using 5 µg of linear DNA. Twenty-four hours after transfection, 32 µg/mL hygromycin B was added for selection of mutants. The L. braziliensis transfectant line was kept for three passages in M199 containing hygromycin B and then plated on semi-solid M199 medium/1% agar supplemented with 1.2 µg/mL biopterin, 2% urine and 32 µg/mL hygromycin B for cloning [24]. Four independent clones were analysed for integration of the cassette into the rDNA locus by PCR amplification, using primers complementary to sequences inside and outside of the linear cassette. Genomic DNA of these clones was purified as described [25] for confirmation by PCR. Primers used were S1 (5'-GATCTGGTTGATTCTGCCAG-3') and S4

(5'-GATCCAGCTGCAGGTTCACC-3') [26] that anneal to the SSU rDNA sequence flanking the insertion sites and primers luc2*F* (5'-GCGGGATCCATGGAAGATGCCAAAAACAT TAAG-3'), luc2*R* (5'-CACGCGCATACATTCACGGCGTTACACGGCGATCTTGCCGC-3') and luc2*i* (5'-GACCGACTACCAGGGCTTCC-3') that anneal to the *luc2* gene contained in the linear cassette (Fig 1A) [22].

Parasite susceptibility to drugs in vitro

Susceptibility of *L. braziliensis* wild type and transgenic lines to amphotericin B and MF was evaluated by a [3-(4,5-dimethylthiazol-2-yl)-2,5-diphenyl tetrazolium bromide] (MTT, Sigma-Aldrich) viability test assay as described [27]. For this assay, promastigotes $(2\times10^6$ parasites per well) were incubated in the presence of increasing concentrations of amphotericin B (0.037 to 0.3 μ M) or MF (25 to 200 μ M) for 24 h. MTT cleavage was measured in a microplate reader (POLARstar Omega, BMG Labtech, Ortenberg, Germany) with a test wavelength of 595 nm and a reference wavelength of 690 nm.

For intracellular amastigotes, bone marrow-derived macrophages (BMDM) from BALB/c mice were used as previously described [21]. BMDM were plated on round glass coverslips in 24-well culture dishes, at a density of 3×10^5 cells in 400 µL of RPMI 1640 medium (Gibco, Invitrogen Corporation) supplemented with 10% FCS (Gibco, Invitrogen Corporation) in a 5% CO₂ atmosphere for 16 h at 37°C allowing macrophages to adhere. Macrophages were then infected with stationary-phase promastigotes using ratios varying from 20 to 35 parasites per macrophage for 3 h at 33°C. Non-internalized parasites were removed by washing with warmed PBS, followed by the addition of fresh medium containing increasing concentrations of amphotericin B (0.025 to 0.2 µM) or MF (0.5 to 24 µM). After 48 h, the cells were fixed in methanol and stained using the panoptic haematological labelling method (Instant Prov kit, Newprov, Pinhais, PR, Brazil). The percentage of infected macrophages was determined by counting 100 cells in three independent experiments. The half-maximal effective concentrations (EC₅₀) were determined from sigmoidal regression of the concentration-response curves. Assays were performed in triplicate and results are expressed as the mean and standard deviation (SD) of at least three independent experiments.

Luciferase assay in vitro

The activity of luciferase was measured in recombinant *L. braziliensis* promastigotes at logarithmic and stationary growth phases. Parasites were washed twice with PBS and then serially diluted for the luciferase assay, which was performed using the One Glo Luciferase Assay System (Promega Corporation) according to the manufacturer's instructions and as described previously [22]. Bioluminescence was measured in a microplate reader (POLARstar Omega, BMG Labtech, Ortenberg, Germany). Each point was tested in triplicate in two independent experiments.

Animal infections

Male golden hamsters (3 to 5 weeks-old) were obtained from the Instituto de Medicina Tropical, University of São Paulo. Female BALB/c mice (3 to 5 weeks-old) were from the Instituto de Ciências Biomédicas, University of São Paulo. Animals were kept in cages and received unlimited food and water.

Given that the wild type (*Lb*-WT) and luciferase transgenic parasites (*Lb*-LUC) were in different passages as promastigote cultures, they were initially used to inoculate highly susceptible hamsters. Initially, 10^6 stationary phase promastigotes were injected into hamsters' hind footpads in a volume of 30 µL. Once the infections were established, parasites of both lines were

Α 5' SSU SAS luc2 a-tub IR hyg CPB IR 3' SSU 1.6 kb ⊣*luc2*R luc2F ⊢ 2.4 kb S1 L---------| luc2R 6.7 kb luc2i | С В D luc2F/luc2R S1/luc2R luc2i/S4 2 2 3 4 5 1 2 1 1 15 Kb RLU (x10⁵) 10 $r^2 = 0.9919$ 9.4 Kb 6.5 5 2.3 2.0 - 6.7 kb 10 20 30 40 50 2.4 kb 2.3 Number of promastigotes (x10⁵) 20

Fig 1. Characterisation of a transgenic line of *L. braziliensis* expressing luciferase. (A) Schematic representation of the linear cassette for integration in the SSU rDNA *locus*. SSU, small subunit; SAS, splice acceptor site; *luc2*, *luc2* coding sequence; α -tub IR, intergenic region of the *L. enrietii* α -tubulin gene; *hyg*, hygromycin phosphotransferase gene; CPB IR, *L. mexicana* cysteine protease B gene intergenic region. Position of the primers and size of amplified fragments are indicated by the dotted lines. (B) Size separation of PCR products amplified using primers complementary to the *luc2* ORF extremities from genomic DNA purified from four independent transfected clones (lanes 1 to 4) and from the wild type strain (lane 5). (C) Agarose gel electrophoresis of amplified products obtained from *Lb*-WT (lane 1) or *Lb*-LUC (lane 2) with the pairs of primers S1/luc2R or luc2i/S4. (-) indicates the control reaction without template DNA. (D) Luminescence intensity and number of *Lb*-LUC promastigotes. Promastigotes were serially diluted and luminescence was measured using a microplate reader, after addition of luciferin. RLU, relative light units.

doi:10.1371/journal.pntd.0004660.g001

recovered from the inoculated footpads and differentiated back into promastigotes in M199 at 25°C. These promastigotes were then used to inoculate BALB/c mice using the same protocol described above, except for the host. When the lesions developed, parasites were recovered, transformed into promastigotes and used to inoculate mice and hamsters with the results described in the Results section. Disease progression was followed for 11 weeks in hamsters and for 24 weeks in BALB/c mice.

Treatment of mice with MF and evaluation of treatment efficacy

BALB/c mice inoculated with *Lb*-WT or *Lb*-LUC were submitted to treatment with MF. Treatment was initiated 4 weeks post-inoculation, using two different doses of the drug: 5 or 15 mg/kg/day, in groups of 5 animals. MF was prepared daily from a stock solution of 6 mg/mL and was administered by oral gavage for 15 consecutive days. An untreated group was used as a control for each line studied. Lesion sizes were evaluated once a week by measuring the difference in thickness between the infected and the contralateral uninfected footpad using a caliper (Mitutoyo Corporation, Kawasaki, Kanagawa, Japan).

Parasite burden in *Lb*-LUC inoculated BALB/c mice was determined 6 weeks (at the end of MF treatment), 17 and 23 weeks post-inoculation. The parasite load was quantified by measuring luciferase activity through bioimaging (IVIS Spectrum, Caliper Life Sciences, Inc. MA/USA) as described [21]. Briefly, before luminescence detection animals received 75 mg/kg luciferin (VivoGlo Luciferin, Promega) intraperitoneally followed by anesthesia in a 2% isoflurane atmosphere (Cristália). Images were collected after 20 minutes using a high-resolution mode with 2 minutes exposure from a fixed-size region of interest. Results were quantified with Living Image software version 4.3.1 (Caliper Life Sciences) and were expressed as photons/second/square centimeter/steradian (ph/sec/cm²/sr).

Susceptibility to MF after in vivo treatment

Lb-WT parasites were recovered from infected mice that had been treated with 5 mg/kg/day (n = 5) or with 15 mg/kg/day (n = 5) MF at the 19th and 23th week post-inoculation, respectively. These parasites were differentiated into promastigotes in M199 medium and their susceptibility to MF was determined by the MTT assay as described above. As controls, parasites were also rescued from an untreated infected animal at the 19th week post-infection.

Statistical analysis

Unpaired two-tailed Student's *t* tests were used to compare the EC₅₀ determined *in vitro* on promastigotes and intracellular amastigotes of wild type parasites and luciferase expressing lines and for comparing disease progression. Parasite burden was analysed for statistical significance by One Way ANOVA, followed by the Tukey post-test. All statistical analyses were performed using GraphPad Prism 6.0 software (GraphPad Software, Inc., La Jolla, CA USA). Results were considered significant at P < 0.05.

Ethics statement

Animal experiments were approved by the Ethics Committee for Animal Experimentation of the Biomedical Sciences Institute (Protocol: 178/138/02) and of the Tropical Medicine Institute (Protocol: CPE-IMT 2012/145) of the University of São Paulo in agreement with the guidelines of the Sociedade Brasileira de Ciência de Animais de Laboratório (SBCAL) and of the Conselho Nacional de Controle da Experimentação Animal (CONCEA).

Results

Generation and characterisation of a luciferase expressing line of *L*. *braziliensis*

To generate a luciferase expressing *L. braziliensis* line (*Lb*-LUC), promastigotes of the H3227 strain were transfected with a linear cassette containing the *luc2* gene flanked by SSU rDNA sequences (Fig 1A). After transfection, parasites were selected in the presence of hygromycin B and cloned. Four independent clones were tested for the presence of the *luc2* gene by PCR. Amplification of a 1.6 kb fragment with oligonucleotides complementary to the 5' and 3' ends of the *luc2* open reading frame was observed in the four selected clones while the same reaction led to the absence of amplification from wild type parasites (Fig 1B). Additional pairs of primers were used to confirm the correct integration into the ribosomal DNA *locus* in one of the selected clones (clone 3, lane 3 in Fig 1B). Primers S1 and S4 [26], complementary to the rDNA outside the linear cassette, were used in combination with primers luc2*R* and luc2*i*, respectively (Fig 1A). These two combinations of primers amplified the expected fragments of 2.4 and 6.7 kb in the transgenic line but not in wild type parasites (Fig 1A and 1C).

Bioluminescence in the transgenic line (*Lb*-LUC) was confirmed after incubation of parasites with luciferin. A linear correlation between the number of promastigotes and the production of light was observed (Fig 1D).

Growth curves for transgenic *Lb*-LUC and wild type (*Lb*-WT) promastigotes were indistinguishable (<u>S1A Fig</u>). Logarithmic and stationary phase promastigotes presented similar light emission (<u>S1B Fig</u>) indicating that luciferase expression was stable along the growth curve. The stability of the transgene was also evaluated by removing the drug pressure from promastigote cultures. Luciferase expression was stable with the same levels of light production in parasites kept in the absence of hygromycin B for 25 passages in culture (<u>S1C Fig</u>).

No differences were observed in the infectivity of *Lb*-LUC and *Lb*-WT to macrophages after 24 or 48 hours evaluated by the morphology of infected cells, percentage of infected cells or number of amastigotes per infected macrophage (S2A–S2D Fig). In addition, a direct correlation was observed between the number of *Lb*-LUC intracellular amastigotes per macrophage and bioluminescence (S2E Fig). Finally, *in vitro* susceptibility of *Lb*-LUC and *Lb*-WT promastigotes and intracellular amastigotes to amphotericin B and MF showed no significant differences (Table 1).

Disease progression in hamsters and BALB/c mice inoculated with *Lb*-LUC

Hamsters and BALB/c mice were inoculated with 10^{6} Lb-WT or Lb-LUC promastigotes subcutaneously in the footpad. In both animal models, there were no significant differences in the kinetics of lesion development or in lesion size when the two parasite lines were compared (Fig 2A and 2B). In both models, lesions were non-ulcerative (Fig 2C and 2D). In hamsters, lesions were detectable from two weeks post-inoculation, progressed steadily for the next five weeks and then the oedema stabilized till the 11th week post-inoculation (Fig 2A), when animals were euthanized. In BALB/c mice, lesions were clearly detected after two weeks, progressed in size until weeks 9–10 and then decreased in the following weeks (Fig 2B and S3 Fig), but the local oedema remained detectable until 23 weeks post-inoculation (Fig 2B and S3 Fig). Therefore, no spontaneous healing was observed in the animal models studied during the period evaluated.

BALB/c mice inoculated with *Lb*-LUC were examined by *in vivo* imaging after 6, 17 and 23 weeks (Figs <u>3</u> and <u>4</u>). Luminescence was detected from the lesion site in all mice at the 6th week post-inoculation (Figs <u>3B</u> and <u>4</u>). By week 17th, light emission was not detected in 2 out of 5 mice (Figs <u>3C</u> and <u>4</u>). No bioluminescence was detected in untreated animals at the 23th week post-inoculation (Figs <u>3D</u> and <u>4</u>), although measurable oedema was still observed in this group at the 24th week (Fig <u>3A</u>).

In vivo efficacy of MF in Lb-LUC inoculated BALB/c mice

The late-healing infection model of *L. braziliensis* H3227 in BALB/c mice was used to test MF efficacy. Treatment was initiated 4 weeks post-inoculation in animals injected with the transgenic line *Lb*-LUC (Fig 3) or with the wild type parasites (S3 Fig). Animals were treated with MF 5 or 15 mg/kg/day for 15 consecutive days. Both doses of MF resulted in regression of lesions at the end of the treatment (6 weeks post-inoculation) (Fig 3A and S4 Fig). Luminescence was not detected by bioimaging at the end of the treatment (Figs <u>3B</u> and <u>4</u>).

However, 8 weeks post-inoculation a relapse was observed in animals treated with 5 mg/kg/ day MF, with lesions progressing in size for the following weeks (Fig 3A and S3 Fig). These findings were confirmed by quantification of bioluminescence 120 days post-inoculation (17^{th} week), time when animals treated with 5 mg/kg/day presented a higher parasite burden than untreated animals (Figs <u>3C</u> and <u>4</u>).

Table 1. Drug susceptibility in promastigotes and intracellular amastigotes of the wild type and the transgenic line of L. braziliensis H3227.

Drug	EC ₅₀ ± S.D. ^a			
	Promastigotes		Intracellular amastigotes	
	Lb-WT	Lb-LUC	Lb-WT	Lb-LUC
amphotericin B	0.11 ± 0.01	0.11 ± 0.02	0.05 ± 0.02	0.05 ± 0.01
miltefosine	40.70 ± 8.55	47.71 ± 5.85	0.99 ± 0.05	0.91 ± 0.01

^a Half maximal effective concentration ± standard deviation (in µM). Results are the average of at least three independent experiments.

doi:10.1371/journal.pntd.0004660.t001



Fig 2. Course of disease after *L. braziliensis* H3227 inoculation in hamsters and BALB/c mice. Evolution of lesion size in hamsters (A) and BALB/c mice (B) inoculated with *Lb*-LUC or *Lb*-WT H3227 strain. Animals were inoculated with 10^6 stationary-phase promastigotes in the left hind footpad and lesion size was measured weekly (five animals per group). Data is the mean and standard deviation from one experiment representative of two independent experiments. Student's *t* test comparing the area under the curve detected non-statistically significant differences with P = 0.25 (A) and P = 0.11 (B). Representative images of a hamster at the 5th week (C) and of a BALB/c mouse at the 6th week post-inoculation (D) with H3227 *Lb*-LUC parasites. The left footpad (inoculated) and right footpad (control) of a hamster (C) and a BALB/c mouse (D), representative of each group of animals, are shown.

doi:10.1371/journal.pntd.0004660.g002





(6th week post-inoculation) (B), at the 17th (C) and at the 23th week post-inoculation (D) (40, 120 and 160 days post-inoculation respectively). Statistical analysis was performed with One Way ANOVA, followed by the Tukey post-test. * p < 0.05; **, p < 0.001. Ph/sec/cm²/sr, photons per second per square centimetre per steradian.

doi:10.1371/journal.pntd.0004660.g003

A clinically detectable relapse was also observed in animals treated with 15 mg/kg/day from the 15th week post-inoculation that progressed in the following weeks (Fig 3A). Although no luminescence was detected in these animals at week 17th (120 days) post-inoculation (Figs <u>3C</u> and <u>4</u>), parasites were clearly detected by *in vivo* imaging at the 23th week post-inoculation (Figs <u>3D</u> and <u>4</u>).

Animals inoculated with *Lb*-WT and treated with both doses of MF presented the same pattern of disease progression, response and relapse to treatment observed with the *Lb*-LUC transgenic line (Fig 3A and S3 Fig).

To investigate whether these relapses were due to MF resistance acquired during the therapy regimen, parasites were recovered from 5 animals treated with MF 5 mg/kg/day (at the 19th week post-inoculation) and with 15 mg/kg/day (at the 23th week post-inoculation) (<u>S3 Fig</u>). After differentiation to promastigotes, MF susceptibility was determined in parasites isolated from untreated and treated animals and no significant changes in drug susceptibility were found between these parasites (<u>Table 2</u>).

Discussion

To the best of our knowledge, this is the first report of a luciferase-expressing *L. braziliensis*, the causative species of cutaneous and mucocutaneous leishmaniasis in humans and the most prevalent species in South America [2]. The importance of research on *L. braziliensis* infections cannot be overstated. However, it faces multiple difficulties largely derived from the distinct



Fig 4. Parasite burden of left hind footpad of BALB/c mice treated with MF and evaluated by bioluminescence. Groups of 5 animals untreated (NT), or treated with 5 or 15 mg/kg/day MF (MF 5 and MF 15 respectively) were evaluated at 40, 120 and 160 days (6th, 17th and 23th week respectively) post-inoculation (pi). The bars on the right show a pseudo-colour scale representing light intensities.

doi:10.1371/journal.pntd.0004660.g004

Treatment regimen	<i>Lb</i> -WT lines ^a	EC ₅₀ ± S.D. ^b
Untreated	<i>Lb</i> -WT (NT.1)	40.88 ± 6.77
5 mg/kg/day	<i>Lb</i> -WT (MF5.1)	36.82 ± 6.10
	<i>Lb</i> -WT (MF5.2)	34.33 ± 8.33
	<i>Lb</i> -WT (MF5.3)	38.12 ± 9.75
	<i>Lb</i> -WT (MF5.4)	41.45 ± 3.80
	<i>Lb</i> -WT (MF5.5)	43.26 ± 9.05
15 mg/kg/day	Lb-WT (MF15.1)	28.24 ± 2.31
	Lb-WT (MF15.2)	28.20 ± 4.46
	Lb-WT (MF15.3)	30.80 ± 4.76
	Lb-WT (MF15.4)	29.00 ± 9.78
	<i>Lb</i> -WT (MF15.5)	36.34 ± 8.68

Table 2. Susceptibility of promastigotes to MF after recovery from BALB/c mice treated with 5 or 15 mg/kg/day.

^a Amastigotes were rescued from *Lb*-WT inoculated BALB/c mice treated with MF 5 mg/kg/day (5 animals), 15 mg/kg/day (5 animals) or untreated (one animal) (MF5, MF15 and NT.1 respectively), differentiated in M199 to promastigotes and then tested for MF susceptibility by MTT as described in Methods.
^b Half maximal effective concentration ± standard deviation (in μM). Results are the average of at least three independent experiments.

doi:10.1371/journal.pntd.0004660.t002

behaviour of this *Leishmania* species, *in vitro* and *in vivo*. *L. braziliensis* promastigotes do not grow easily in axenic cultures and amastigotes are scarce in the lesions. The lack of an appropriate animal model mimicking the human infection in all of its potential clinical presentations is perhaps one of the main drawbacks. Most *L. braziliensis* strains are infective to mice but lead to asymptomatic or short-lived, self-healing disease [28, 29]. The infection is more sustained in hamsters [6, 29], but do not progress to a mucosal or disseminated disease.

The development of lesions at the site of inoculation with the H3227 strain of *L. braziliensis* in BALB/c mice was previously described [10]. However, in the original description, lesions were self-limited and healed 30 days post-inoculation. This strain was kindly given to us by Dr. Maria Jania Teixeira (Universidade Federal do Ceará, Brazil) and, before being submitted to transfection, was passaged in mice repeatedly. Most likely as a result of these repeated *in vivo* passages, the line obtained after transfection, together with the parental wild type parasites were then able to induce a late-healing disease in BALB/c mice, as well as being pathogenic in hamsters. A late-healing animal model of *L. braziliensis* infection is likely to become very useful as a tool in drug development or in the evaluation of the disease pathogenesis.

It is unclear at present whether other strains of *L. braziliensis* could be adapted to cause latehealing disease in BALB/c mice. We have attempted to do so with three other isolates as well as with the type strain of *L. braziliensis* M2903 without success but cannot rule out that possibility. It will be interesting to investigate whether mutations on this *Leishmania* isolate (H3227) are present and could be correlated with the altered phenotype.

These parasites were transformed to express the *luciferase* gene integrated into the ribosomal *locus*. Luciferase expression was stable throughout the life cycle. As previously reported for other luciferase-expressing *Leishmania* species [21, 22, 30], this *L. braziliensis* transgenic line presented the same biological properties *in vitro* and *in vivo* of the parental line and was useful for parasite quantification of promastigotes and amastigotes, *in vitro* and *in vivo*. Luciferase as a reporter has become instrumental in pre-clinical drug development projects, especially in chronic infections. Our previous results with *L. amazonensis* [21] and *L. chagasi* [22] lines expressing luciferase clearly demonstrated that luciferase detection correlates well with parasite burden *in vivo*. In the case of H3227 parasites, the sensitivity of detection *in vivo* has proven to be insufficient to detect low parasite burdens, as was the case in *Lb*-LUC in untreated animals or when relapses were noticed in the groups treated with the higher dose of MF. This limitation in sensitivity should be taken into account in drug efficacy studies. Nevertheless, bioimaging was used successfully to document infection and as a semi-quantitative method for comparing parasite burden between different groups.

The model of infection with *Lb*-LUC in BALB/c mice was put to test using MF, the newest addition to leishmaniasis chemotherapy. The susceptibility of the H3227 strain to MF *in vitro* can be considered comparable to other *L. braziliensis* strains [31, 32]. Accordingly, at the end of MF treatment, *Lb*-LUC inoculated mice showed regression of lesions and absence of luminescence. However, although the doses used here were higher than the WHO recommended scheme for CL and VL (2.5 mg/kg/day) [15], a clinical relapse was evident in animals treated with 5 mg/kg/day as soon as two weeks after the end of treatment. In these animals, lesions progressed in the following weeks reaching sizes similar to the ones observed in untreated animals. In animals treated with 15 mg/kg/day, oedema in the footpad reappeared 10 weeks after the interruption of treatment.

Relapses in VL patients treated with MF do not seem to be due to acquired resistance [33]. Likewise, our data showed similar MF susceptibility in parasites rescued from untreated and treated animals, indicating that *L. braziliensis* did not acquire resistance to MF during treatment.

In VL patients, MF failure may be correlated with low drug exposure due to pharmacokinetic-pharmacodynamics factors [34]. MF has been shown to reach high levels in the skin [15] but poor drug distribution to cutaneous lesions may be another reason for the lower efficacy of the drug against CL when compared with VL. Furthermore, the greater genetic diversity of *L*. *braziliensis* compared to *L*. *donovani* [35, 36] may increase the number of elements that contribute to treatment failure. Moreover, other factors, unrelated to drug resistance, should be considered as quiescence and even the presence of the LRV1 virus [37, 38].

The persistence of viable parasites after treatment with 5 and 15 mg/kg/day MF per 15 days indicated that, even for the highest dose used, clinical response did not reflect sterile cure. In fact, sterile cure does not seem to be required for long-term control of the disease, as has been shown by the presence of parasites in healthy skin in animals [39, 40]. On the other hand, relapses have been reported in patients treated with MF for all clinical forms of the disease [33, 41-46]. Therefore, we believe the data presented herein stress the need for long-term follow up of MF treated patients and suggest that higher doses and/or longer treatment regimens should be evaluated in order to avoid relapses. In addition, the combined use of MF with other antileishmanial drugs should be considered as an alternative for the chemotherapy of the disease.

Supporting Information

S1 Fig. Characterization of the transgenic line of *L. braziliensis* expressing luciferase. (A) Growth curves for *Lb*-WT and *Lb*-LUC promastigote lines. Parasites were grown in M199 and then counted using a Neubauer hemocytometer. Data is the mean and standard error from two independent experiments. RLU, relative luminescence units. (B) Luciferase activity in logarithmic and stationary-phase *Lb*-LUC promastigotes. (C) Stability of luciferase expression in *Lb*-LUC. Promastigotes were cultivated for 25 passages in the presence or absence of hygromycin B and then tested for luminescence. (DOCX)

S2 Fig. *In vitro* infectivity of *Lb*-LUC in BMDM. Macrophages were infected with stationary phase promastigotes for 3 hours at 33°C. Non-internalized parasites were removed by washing with warmed PBS. (A) and (B): Representative images of macrophages infected with *Lb*-WT (A) and *Lb*-LUC (B). (C) Percentage of infected macrophages after 24 or 48 hours. Cells were infected with a ratio of 20 parasites per macrophage. The percentage of infected macrophage after 24 or 48 hours. Cells in triplicates. (D) Number of amastigotes per infected macrophage after 24 or 48 hours. (E) Macrophage infection evaluated through luminescence. Relative luminescence units (RLU) were determined for macrophages after infection with different ratios of parasites per macrophage (25:1, 30:1 and 35:1). The average and standard deviation of three independent experiments in triplicates is shown for both experiments. Anova with Tukey's post test: (*) *P*< 0.05; (**) *P*<0.01; (***) *P*<0.001. (DOCX)

S3 Fig. *In vivo* efficacy of miltefosine in *L. braziliensis*-inoculated BALB/c mice. 10^6 *Lb*-WT stationary-phase promastigotes were injected in the mice left hind footpad. Treatment with 5 or 15 mg/kg/day of miltefosine was given for 15 consecutive days (from 4th to 6th week post-inoculation). Average of lesion size was measured weekly (five animals per group) in untreated and treated animals The horizontal black bar indicates MF treatment and the arrows indicate the time points when parasites were recovered from BALB/c mice treated with 5 or 15 mg/kg/ day (19th and 23th week respectively). (DOCX)

Acknowledgments

We thank Dr. Maria Jania Teixeira, from Universidade Federal do Ceará for providing the parasites and Dr. Tony Aebischer and Dr. Marc Ouellette for providing plasmids used in this study. We also thank Jenicer K. U. Yokoyama-Yasunaka for technical assistance.

Author Contributions

Conceived and designed the experiments: ACC CTT SRBU. Performed the experiments: ACC JCO CRE JQR CTT. Analyzed the data: ACC JCO CTT SRBU. Wrote the paper: ACC JQR CTT SRBU.

References

- Alvar J, Velez ID, Bern C, Herrero M, Desjeux P, Cano J, et al. Leishmaniasis Worldwide and Global Estimates of Its Incidence. PLoS One. 2012; 7(5):e35671. Epub 2012/06/14. doi: <u>10.1371/journal.pone.0035671</u> PMID: <u>22693548</u>; PubMed Central PMCID: PMC3365071.
- Murray HW, Berman JD, Davies CR, Saravia NG. Advances in leishmaniasis. Lancet. 2005; 366 (9496):1561–77. Epub 2005/11/01. doi: <u>10.1016/s0140-6736(05)67629-5</u> PMID: <u>16257344</u>.
- Llanos Cuentas EA, Cuba CC, Barreto AC, Marsden PD. Clinical characteristics of human *Leishmania* braziliensis braziliensis infections. Trans R Soc Trop Med Hyg. 1984; 78(6):845–6. Epub 1984/01/01. PMID: <u>6533860</u>.
- Jones TC, Johnson WD Jr., Barretto AC, Lago E, Badaro R, Cerf B, et al. Epidemiology of American cutaneous leishmaniasis due to *Leishmania braziliensis* braziliensis. J Infect Dis. 1987; 156(1):73–83. Epub 1987/07/01. PMID: <u>3598227</u>.
- Jirmanus L, Glesby MJ, Guimaraes LH, Lago E, Rosa ME, Machado PR, et al. Epidemiological and clinical changes in American tegumentary leishmaniasis in an area of *Leishmania (Viannia) braziliensis* transmission over a 20-year period. Am J Trop Med Hyg. 2012; 86(3):426–33. Epub 2012/03/10. doi: <u>10.4269/ajtmh.2012.11-0378</u> PMID: <u>22403312</u>; PubMed Central PMCID: PMCPMC3284357.
- Gomes-Silva A, Valverde JG, Ribeiro-Romao RP, Placido-Pereira RM, Da-Cruz AM. Golden hamster (*Mesocricetus auratus*) as an experimental model for *Leishmania (Viannia) braziliensis* infection. Parasitology. 2013; 140(6):771–9. Epub 2013/02/02. doi: <u>10.1017/s0031182012002156</u> PMID: <u>23369503</u>.

- Neal RA, Hale C. A comparative study of susceptibility of inbred and outbred mouse strains compared with hamsters to infection with New World cutaneous leishmaniases. Parasitology. 1983; 87 (Pt 1):7– 13. Epub 1983/08/01. PMID: <u>6622066</u>.
- DeKrey GK, Lima HC, Titus RG. Analysis of the immune responses of mice to infection with *Leishmania braziliensis*. Infect Immun. 1998; 66(2):827–9. Epub 1998/02/07. PMID: <u>9453649</u>; PubMed Central PMCID: PMCPmc107977.
- Rocha FJ, Schleicher U, Mattner J, Alber G, Bogdan C. Cytokines, signaling pathways, and effector molecules required for the control of *Leishmania (Viannia) braziliensis* in mice. Infect Immun. 2007; 75 (8):3823–32. Epub 2007/05/23. doi: <u>10.1128/iai.01335-06</u> PMID: <u>17517868</u>; PubMed Central PMCID: PMCPmc1951993.
- Indiani de Oliveira C, Teixeira MJ, Teixeira CR, Ramos de Jesus J, Bomura Rosato A, Santa da Silva J, et al. *Leishmania braziliensis* isolates differing at the genome level display distinctive features in BALB/c mice. Microbes Infect. 2004; 6(11):977–84. Epub 2004/09/04. doi: <u>10.1016/j.micinf.2004.05.009</u> PMID: <u>15345228</u>.
- Bryceson A. A policy for leishmaniasis with respect to the prevention and control of drug resistance. Trop Med Int Health. 2001; 6(11):928–34. Epub 2001/11/13. 795 [pii]. PMID: <u>11703848</u>.
- Croft SL, Sundar S, Fairlamb AH. Drug resistance in leishmaniasis. Clin Microbiol Rev. 2006; 19 (1):111–26. Epub 2006/01/19. 19/1/111 [pii] doi: <u>10.1128/CMR.19.1.111-126.2006</u> PMID: <u>16418526</u>; PubMed Central PMCID: PMC1360270.
- Jha TK, Sundar S, Thakur CP, Bachmann P, Karbwang J, Fischer C, et al. Miltefosine, an oral agent, for the treatment of Indian visceral leishmaniasis. N Engl J Med. 1999; 341(24):1795–800. Epub 1999/ 12/10. doi: 10.1056/nejm199912093412403 PMID: 10588964.
- Sundar S, Jha TK, Thakur CP, Engel J, Sindermann H, Fischer C, et al. Oral miltefosine for Indian visceral leishmaniasis. N Engl J Med. 2002; 347(22):1739–46. Epub 2002/11/29. doi: <u>10.1056/</u> NEJMoa021556 PMID: 12456849.
- Dorlo TP, Balasegaram M, Beijnen JH, de Vries PJ. Miltefosine: a review of its pharmacology and therapeutic efficacy in the treatment of leishmaniasis. J Antimicrob Chemother. 2012; 67(11):2576–97. dks275 [pii] doi: 10.1093/jac/dks275 PMID: 22833634.
- Soto J, Arana BA, Toledo J, Rizzo N, Vega JC, Diaz A, et al. Miltefosine for new world cutaneous leishmaniasis. Clin Infect Dis. 2004; 38(9):1266–72. Epub 2004/05/06. doi: <u>10.1086/383321</u> PMID: 15127339.
- Soto J, Berman J. Treatment of New World cutaneous leishmaniasis with miltefosine. Trans R Soc Trop Med Hyg. 2006; 100 Suppl 1:S34–40. Epub 2006/08/26. doi: <u>10.1016/j.trstmh.2006.02.022</u> PMID: <u>16930649</u>.
- Rubiano LC, Miranda MC, Muvdi Arenas S, Montero LM, Rodriguez-Barraquer I, Garcerant D, et al. Noninferiority of miltefosine versus meglumine antimoniate for cutaneous leishmaniasis in children. J Infect Dis. 2012; 205(4):684–92. Epub 2012/01/13. doi: <u>10.1093/infdis/jir816</u> PMID: <u>22238470</u>; PubMed Central PMCID: PMCPMC3266136.
- Chrusciak-Talhari A, Dietze R, Chrusciak Talhari C, da Silva RM, Gadelha Yamashita EP, de Oliveira Penna G, et al. Randomized controlled clinical trial to access efficacy and safety of miltefosine in the treatment of cutaneous leishmaniasis Caused by *Leishmania (Viannia) guyanensis* in Manaus, Brazil. Am J Trop Med Hyg. 2011; 84(2):255–60. Epub 2011/02/05. doi: <u>10.4269/ajtmh.2011.10-0155</u> PMID: <u>21292895</u>; PubMed Central PMCID: PMC3029178.
- Machado PR, Ampuero J, Guimaraes LH, Villasboas L, Rocha AT, Schriefer A, et al. Miltefosine in the treatment of cutaneous leishmaniasis caused by *Leishmania braziliensis* in Brazil: a randomized and controlled trial. PLoS Negl Trop Dis. 2010; 4(12):e912. Epub 2011/01/05. doi: <u>10.1371/journal.pntd.</u> <u>0000912</u> PMID: <u>21200420</u>; PubMed Central PMCID: PMC3006132.
- Reimao JQ, Trinconi CT, Yokoyama-Yasunaka JK, Miguel DC, Kalil SP, Uliana SR. Parasite burden in Leishmania (Leishmania) amazonensis-infected mice: Validation of luciferase as a quantitative tool. J Microbiol Methods. 2013; 93(2):95–101. Epub 2013/03/08. doi: <u>10.1016/j.mimet.2013.02.007</u> PMID: <u>23466934</u>.
- Reimao JQ, Oliveira JC, Trinconi CT, Cotrim PC, Coelho AC, Uliana SR. Generation of luciferaseexpressing *Leishmania infantum chagasi* and assessment of miltefosine efficacy in infected hamsters through bioimaging. PLoS Negl Trop Dis. 2015; 9(2):e0003556. Epub 2015/02/14. doi: <u>10.1371/journal.</u> <u>pntd.0003556</u> PMID: <u>25679212</u>; PubMed Central PMCID: PMCPMC4332486.
- Arruda DC, D'Alexandri FL, Katzin AM, Uliana SR. Leishmania amazonensis: biosynthesis of polyprenols of 9 isoprene units by amastigotes. Exp Parasitol. 2008; 118(4):624–8. Epub 2007/12/25. doi: <u>10.</u> <u>1016/j.exppara.2007.11.009</u> PMID: <u>18155196</u>.
- Coburn CM, Otteman KM, McNeely T, Turco SJ, Beverley SM. Stable DNA transfection of a wide range of trypanosomatids. Mol Biochem Parasitol. 1991; 46(1):169–79. PMID: 1906580.

- Medina-Acosta E, Cross GA. Rapid isolation of DNA from trypanosomatid protozoa using a simple 'mini-prep' procedure. Mol Biochem Parasitol. 1993; 59(2):327–9. PMID: 8341329.
- Uliana SR, Nelson K, Beverley SM, Camargo EP, Floeter-Winter LM. Discrimination amongst *Leishmania* by polymerase chain reaction and hybridization with small subunit ribosomal DNA derived oligonucleotides. J Eukaryot Microbiol. 1994; 41(4):324–30. Epub 1994/07/01. PMID: <u>8087103</u>.
- Zauli-Nascimento RC, Miguel DC, Yokoyama-Yasunaka JK, Pereira LI, Pelli de Oliveira MA, Ribeiro-Dias F, et al. In vitro sensitivity of *Leishmania (Viannia) braziliensis* and *Leishmania (Leishmania) amazonensis* Brazilian isolates to meglumine antimoniate and amphotericin B. Trop Med Int Health. 2010; 15 (1):68–76. Epub 2009/10/31. doi: 10.1111/j.1365-3156.2009.02414.x PMID: 19874570.
- Childs GE, Lightner LK, McKinney L, Groves MG, Price EE, Hendricks LD. Inbred mice as model hosts for cutaneous leishmaniasis. I. Resistance and susceptibility to infection with *Leishmania braziliensis*, *L. mexicana*, and *L. aethiopica*. Ann Trop Med Parasitol. 1984; 78(1):25–34. Epub 1984/02/01. PMID: 6721612.
- Mears ER, Modabber F, Don R, Johnson GE. A Review: The Current In Vivo Models for the Discovery and Utility of New Anti-leishmanial Drugs Targeting Cutaneous Leishmaniasis. PLoS Negl Trop Dis. 2015; 9(9):e0003889. Epub 2015/09/04. doi: <u>10.1371/journal.pntd.0003889</u> PMID: <u>26334763</u>; PubMed Central PMCID: PMCPMC4559374.
- Roy G, Dumas C, Sereno D, Wu Y, Singh AK, Tremblay MJ, et al. Episomal and stable expression of the luciferase reporter gene for quantifying *Leishmania* spp. infections in macrophages and in animal models. Mol Biochem Parasitol. 2000; 110(2):195–206. PMID: <u>11071276</u>.
- Obonaga R, Fernandez OL, Valderrama L, Rubiano LC, Castro Mdel M, Barrera MC, et al. Treatment failure and miltefosine susceptibility in dermal leishmaniasis caused by *Leishmania* subgenus *Viannia* species. Antimicrob Agents Chemother. 2014; 58(1):144–52. Epub 2013/10/23. doi: <u>10.1128/aac.</u> 01023-13 PMID: 24145529; PubMed Central PMCID: PMCPMC3910710.
- de Morais-Teixeira E, Damasceno QS, Galuppo MK, Romanha AJ, Rabello A. The in vitro leishmanicidal activity of hexadecylphosphocholine (miltefosine) against four medically relevant *Leishmania* species of Brazil. Mem Inst Oswaldo Cruz. 2011; 106(4):475–8. Epub 2011/07/09. PMID: <u>21739037</u>.
- Rijal S, Ostyn B, Uranw S, Rai K, Bhattarai NR, Dorlo TP, et al. Increasing failure of miltefosine in the treatment of Kala-azar in Nepal and the potential role of parasite drug resistance, reinfection, or noncompliance. Clin Infect Dis. 2013; 56(11):1530–8. Epub 2013/02/22. doi: <u>10.1093/cid/cit102</u> PMID: 23425958.
- Dorlo TP, Rijal S, Ostyn B, de Vries PJ, Singh R, Bhattarai N, et al. Failure of miltefosine in visceral leishmaniasis is associated with low drug exposure. J Infect Dis. 2014; 210(1):146–53. Epub 2014/01/ 21. doi: <u>10.1093/infdis/jiu039</u> PMID: <u>24443541</u>.
- Cupolillo E, Brahim LR, Toaldo CB, De Oliveira-Neto MP, De Brito ME, Falqueto A, et al. Genetic Polymorphism and Molecular Epidemiology of *Leishmania (Viannia) braziliensis* from Different Hosts and Geographic Areas in Brazil. J Clin Microbiol. 2003; 41(7):3126–32. PMID: 12843052.
- Oliveira FS, Valete-Rosalino CM, Pacheco SJ, Costa FA, Schubach AO, Pacheco RS. American tegumentary leishmaniasis caused by *Leishmania (Viannia) braziliensis*: assessment of parasite genetic variability at intra- and inter-patient levels. Parasit Vectors. 2013; 6:189. Epub 2013/06/22. doi: <u>10.1186/</u><u>1756-3305-6-189</u> PMID: <u>23786878</u>; PubMed Central PMCID: PMCPmc3729673.
- Adaui V, Lye LF, Akopyants NS, Zimic M, Llanos-Cuentas A, Garcia L, et al. Association of the Endobiont Double-Stranded RNA Virus LRV1 With Treatment Failure for Human Leishmaniasis Caused by *Leishmania braziliensis* in Peru and Bolivia. J Infect Dis. 2015. Epub 2015/07/01. doi: <u>10.1093/infdis/</u> jiv354 PMID: <u>26123565</u>.
- Vanaerschot M, Dumetz F, Roy S, Ponte-Sucre A, Arevalo J, Dujardin JC. Treatment failure in leishmaniasis: drug-resistance or another (epi-) phenotype? Expert Rev Anti Infect Ther. 2014; 12(8):937– 46. Epub 2014/05/08. doi: 10.1586/14787210.2014.916614 PMID: 24802998.
- 39. Brandao-Filho SP, Brito ME, Carvalho FG, Ishikawa EA, Cupolillo E, Floeter-Winter L, et al. Wild and synanthropic hosts of *Leishmania (Viannia) braziliensis* in the endemic cutaneous leishmaniasis locality of Amaraji, Pernambuco State, Brazil. Trans R Soc Trop Med Hyg. 2003; 97(3):291–6. Epub 2004/07/ 02. PMID: <u>15228244</u>.
- 40. de Oliveira GM, de Fatima Madeira M, Oliveira FS, Pacheco RS. PCR associated with molecular hybridization detects *Leishmania (Viannia) braziliensis* in healthy skin in canine tegumentary leishmania-sis. J Parasitol. 2015; 101(1):91–3. Epub 2014/08/27. doi: <u>10.1645/14-567.1</u> PMID: <u>25158080</u>.
- Calvopina M, Gomez EA, Sindermann H, Cooper PJ, Hashiguchi Y. Relapse of new world diffuse cutaneous leishmaniasis caused by *Leishmania (Leishmania) mexicana* after miltefosine treatment. Am J Trop Med Hyg. 2006; 75(6):1074–7. Epub 2006/12/19. PMID: <u>17172368</u>.

- Soto J, Toledo J, Valda L, Balderrama M, Rea I, Parra R, et al. Treatment of Bolivian mucosal leishmaniasis with miltefosine. Clin Infect Dis. 2007; 44(3):350–6. Epub 2007/01/06. doi: <u>10.1086/510588</u> PMID: <u>17205440</u>.
- Zerpa O, Ulrich M, Blanco B, Polegre M, Avila A, Matos N, et al. Diffuse cutaneous leishmaniasis responds to miltefosine but then relapses. The Brit J Dermatol. 2007; 156(6):1328–35. Epub 2007/04/ 20. doi: 10.1111/j.1365-2133.2007.07872.x PMID: 17441955.
- Pandey BD, Pandey K, Kaneko O, Yanagi T, Hirayama K. Relapse of visceral leishmaniasis after miltefosine treatment in a Nepalese patient. Am J Trop Med Hyg. 2009; 80(4):580–2. Epub 2009/04/07. PMID: 19346379.
- Velez I, Lopez L, Sanchez X, Mestra L, Rojas C, Rodriguez E. Efficacy of miltefosine for the treatment of American cutaneous leishmaniasis. Am J Trop Med Hyg. 2010; 83(2):351–6. Epub 2010/08/05. doi: <u>10.4269/ajtmh.2010.10-0060</u> PMID: <u>20682881</u>; PubMed Central PMCID: PMCPMC2911184.
- 46. Ghosh S, Das NK, Mukherjee S, Mukhopadhyay D, Barbhuiya JN, Hazra A, et al. Inadequacy of 12-Week Miltefosine Treatment for Indian Post-Kala-Azar Dermal Leishmaniasis. Am J Trop Med Hyg. 2015. Epub 2015/07/16. doi: <u>10.4269/ajtmh.14-0721</u> PMID: <u>26175030</u>.